

## **European Journal of Economics, Law and Politics**

Paper: "The Influence of Observability on The Adoption of Tissue Culture Banana Seedlings in Tanzania"

Submitted: 13 February 2024 Accepted: 11 March 2024 Published: 31 March 2024

Corresponding Author: Tumaini Steven

Email: tumainisteve2018@gmail.com

Doi: 10.19044/elp.v11no1a1

Peer review:

Reviewer 1: Blinded

Reviewer 2: Blinded